Author's response to reviews

Title: Acute Mercury Poisoning: A Case Report

Authors:

Sezgin Sarikaya (sezginsa@hotmail.com)
Ozgur Karciglu (okarol@hotmail.com)
Didem Ay (diedmay@hotmail.com)
Asli Cetin (asliet@hotmail.com)
Can Aktas (canaktas@hotmail.com)
Mustafa Serinen (mserinen@hotmail.com)

Version: 2 Date: 28 September 2009

Author's response to reviews:

To the Editor,
We are now submitting the revised version of our manuscript entitled 'Acute Mercury Poisoning: A Case Report' in accord with the reviewers' recommendations.
We made changes in the manuscript with blue colors.
Correspondence regarding this manuscript may be sent to:
Mustafa Serinen, MD
Pamukkale University Medical School,
Dept. of Emergency Medicine,
20070, Denizli-TURKEY
Tel: +90.505.2991497 (mobile)
Fax: +90.258.2134922
E-mail: mserinen@hotmail.com

Thank you in advance for your time in considering this manuscript.

Sincerely,
Mustafa SERINKEN, MD